Technische Universität München Fakultät für Informatik

Dr. K. N. Verma verma@in.tum.de Room: MI 02.07.041

Language Based Security

Summer Semester 2006

3. Homework 31 May 2006

Exercise 1:

We extend the simple functional language discussed in the lecture (without vectors) by adding division operation. Extend the evaluation and typing rules and show type safety of the language.

Exercise 2:

Extend the URLClassLoader class to have it store a list of bad class names so that loading these classes always fails.